Author's response to reviews

Title: Implementation research design: Integrating participatory action research into randomized controlled trials

Authors:

Luci K Leykum (Leykum@uthscsa.edu)
Jacqueline A Pugh (jacque.pugh@gmail.com)
Holly J Lanham (Holly.Lanham@phd.mccombs.utexas.edu)
Joel Harmon (JHarmon444@aol.com)
Reuben R McDaniel Jr (reuben.mcdaniel@mccombs.utexas.edu)

Version: 6 Date: 12 October 2009

Author's response to reviews:

Thank you for your email dated Thursday October 8th, outlining the final pre-publication formatting changes. I have updated the references to include numbers 18 and 19, and removed all "reviewing text." I am not sure how to edit the tables so that there are no vertical lines, but according to the email, this is something that it is possible for the Implementation Science team to do. Thank you again for your email, and your assistance with this manuscript!

Best regards,

Luci Leykum